Author's response to reviews

Title: Predicting Risk of Hospital and Emergency Department Use for Community Based Elderly Persons through a Secondary Analysis of Cross-National Data

Authors:

John N Morris (jnm@hsl.harvard.edu)
Elizabeth P Howard (e.howard@neu.edu)
Knight Stell (knightrsteel@yahoo.com)
Robert Schreiber (rschreiber@hsl.harvard.edu)
brant E Fries (bfries@umich.harvard.edu)
Lewis A Lipsitz (lipsitz@hsl.harvard.edu)
Beryl Goldman (bgoldman@kendaloutreach.org)

Version: 2 Date: 23 December 2013

Author's response to reviews: see over
Predicting Risk of Hospital and Emergency Department Use for Community-Based Elderly Persons Through a Secondary Analysis of Cross-National Data

John N. Morris, PhD
Hebrew SeniorLife, Institute for Aging Research
1200 Centre Street, Boston, MA 02131
jnm@hsl.harvard.edu

Elizabeth P. Howard, PhD, RN, ACNP
Northeastern University, School of Nursing
360 Huntington Avenue, Boston, MA 02115
e.howard@neu.edu

Knight Steel, MD
Hackensack University Medical Center
30 Prospect Avenue, Hackensack, NJ 07601
knightrsteel@yahoo.com

Robert Schreiber, MD
Hebrew SeniorLife, Institute for Aging Research
1200 Centre Street, Boston, MA 02131
rschreiber@hsl.harvard.edu

Brant E. Fries, PhD
University of Michigan, Institute of Gerontology and Geriatric Research, Education and Clinical Center, Ann Arbor VA Healthcare Center
300 NIB, 933 NW, Ann Arbor, MI 48109
bfries@umich.edu

Lewis A. Lipsitz, MD
Hebrew SeniorLife, Institute for Aging Research
1200 Centre Street, Boston, MA 02131
Lipsitz@hsl.harvard.edu

Beryl Goldman, PhD, RN
Kendal Outreach LLC.
1107 E Baltimore Pike, Kennett Square, PA 19348
bgoldman@kendaloutreach.org
Correspondence concerning this manuscript should be directed to:  John N. Morris